Author's response to reviews

Title: Which field walking test should be used to assess functional exercise capacity in lung cancer? : an observational study

Authors:

Catherine L Granger (catherine.granger@unimelb.edu.au)
Linda Denehy (linda.denehy@unimelb.edu.au)
Selina M Parry (selina.parry@unimelb.edu.au)
Joel Martin (joelmartin39@gmail.com)
Tim Dimitriadis (tim.dimitriadis@mh.org.au)
Maeve Sorohan (maeve_or19@hotmail.co)
Louis Irving (louis.irving@mh.org.au)

Version: 3 Date: 15 July 2015

Author's response to reviews: see over
Dear Dafne Solera

**Re: Manuscript submitted to BMC Cancer titled:** “Which field walking test should be used to assess exercise capacity in lung cancer: an observational study?”

We are delighted to hear that our manuscript has been accepted for submission in your journal. Please find below our response to the editor’s comments.

<table>
<thead>
<tr>
<th>Editor Comments</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Could you provide the name of the ethics committee that approved the study.</td>
<td>The name of the ethics committee that approved the study has been outlined in methods section and now reads: “The study had ethical approval from Melbourne Health Human Research Ethics Committee and written consent was obtained from participants.”</td>
</tr>
<tr>
<td>STROBE: As it is an observational study could you please upload the STROBE checklist and fill in the appropriate information</td>
<td>Please find the STROBE checklist attached to the online submission with relevant sections completed outlining where the information is reported.</td>
</tr>
<tr>
<td>Title: Move the question mark before the colon in the title</td>
<td>The question mark has been moved as requested.</td>
</tr>
<tr>
<td>When you upload the manuscript again could you please remove the tracked change document as an additional file</td>
<td>The tracked changed document has been removed.</td>
</tr>
</tbody>
</table>

Thank you for considering this for publication in your journal. We look forward to your response.

Kind Regards,

**Dr Catherine Granger PhD PT**
Lecturer, Department of Physiotherapy
The University of Melbourne, Australia